Author’s response to reviews

Title: Twenty-five Years On: Revisiting Bosnia and Herzegovina after Implementation of a Family Medicine Development Program.

Authors:

Geoffrey Hodgetts (geoffrey.hodgetts@dfm.queensu.ca)

Glenn Brown (Glenn.Brown@dfm.queensu.ca)

Olivera Batić-Mujanović (oliverabaticmujanovic@yahoo.com)

Larisa Gavran (gavranlarisa@yahoo.com)

Zaim Jatić (jaticzaim@gmail.com)

Maja Račić (porodicnamedicina@gmail.com)

Gordana Tešanović (gordatesan@gmail.com)

Amra Zahilić (azalihic@gmail.com)

Mary Martin (mary.martin@dfm.queensu.ca)

Richard Birtwhistle (richard.birtwhistle@dfm.queensu.ca)

Version: 2 Date: 31 Dec 2019

Author’s response to reviews:

Dec.31, 2019

To: The Editor, BMC FP

Dear Dr. Chenot,

Thank you for your review and suggestions.

1. I would appreciate if you could add a map of BH showing where the Dept. of Family medicine are located and how they are distributed with regard to the complicated borders reflecting the ethnic distributions of the people.

I have added a map showing the entity boundaries and location of Brcko District, with the cities with Departments of Family Medicine marked. This is referred to in the text as Appendix B, in paragraph 3 of Queen’s University Family Medicine Development Program page 4 lines 34-37. This map was created by my son, Paul Hodgetts, who is a graphic designer. Therefore, there are no issues about copyright or permissions from other proprietary sources or use of templates.
available to the public. I have added an acknowledgement to Paul Hodgetts in the Acknowledgement section.

2. Additional Editorial comments:

1. Please reformat figure 1, by removing the box and include the text from the text box as an additional file. Please also ensure that all additional files are explicitly referred to in the main text.

I have removed Figure 1 from the Tables section document and added it as text-only as Appendix A. This is referenced in the last line of Background section on page 3.

2. Please proofread and ensure that when you upload your revised submission it is in the final form for publication. Please remove any tracked changes or highlighting and include only a single clean copy of the manuscript.

The manuscript, Tables and Appendices have been proofread and all track changes removed.

Thank you again for your consideration of this paper and all the helpful changes and suggestions.

Happy New Year!

P. Geoffrey Hodgetts MD CCFP FCFP

Prof. Emeritus, Dept. of Family Medicine

Queen’s University, Kingston, Canada.